vc/12-Sep-03
Week 37 starting 08 Sept 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today(09/09)
	Forward look,   & Next Magnet 

	TBA1
	
	Empty
	CFU REPAIRED BY ACR GROUP – Next Magnet 2012 Expected 09 or 10 Sept.

	TBA2
	3028
	Being Cooled
	 COLd Tests start  by Wednesday  10th? 

	TBC1
	3046
	cold
	After Thermal Cyc Tests & warm up & disconnection by 11th 

	TBC2
	SSS1
	Warm 
	Mech. Mods. To prepare for SSS1 then, Connection of SSS1

	TBE1
	2018
	Disconnected but cannot be removed due to SM18 floor painting
	2018 needs removal & bring 3043 for connection

	TBE2
	1031
	cold
	Cold Tests ; Disconnect on 10th  Bring  New after(2010 ?)

	TBF1
	2025 after F2 done? 
	
	Prepare For 

Commissioning AFTER F2

	TBF2
	2025 proposed
	
	Prepare For 

Commissioning FIRST


· Test Coordinator this week:  P Pugnat     MTA Technician on call:  (16-3602) : Guy

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Portable Phone(s)
· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Monday 1_M


	IAP @warm (TBE1): one amplifier crate was not powered
	MTM
	
	Equip.support called

Solved

	Monday 1_A
	TBA1: TT821 sensor for the 3028: during 4 wires measurement resistance circuit was found to be open.

Circuit open between I+/I-
	MTM
	IAP
	Euip.support called.

Bricolage (shortening of the opened wire in the rack)



	Monday 1_A
	TBC1: Problem in the measurement of the magnetic field using shaft 24 (ap2). 
	MTM
	
	 Stephane, Marco called. Investigation on going. Stephane advised to perform full load line as well.

Problem not resolved.

	Monday 1_A
	TBC1: PLC OK lost during full load line at 7000 A.Work-station blocked.
	CO
	
	 Test aborted and repeated

	Monday 1_N
	MTF transfer file problem: create a new document.
	MTM/EDMS
	
	

	Monday 1_N
	TBC1: minimum energy at 3 kA : launch thermal cycle with no Tcoil
	MTM
	
	.



	Thursday 2_M
	TBA1: HV test (insulation problem Dipole/earth)
	MTM
	
	R.Mompo and equip.support called.

Disconnection of the magnet and test of the CFU.

Problem in the CFU (position of the heaters modified)

Change of bench for the magnet

Same problem in TBA2

Solved now.



	Thursday 2_A
	TBC1: SSL at 4.2 K PLC gives not OK.

Software problem: new version of appli for PLC not updated.
	AB/CO
	
	Resolved: old version loaded.



	Thursday 28_A

Thursday 28_N
	TBC : faulty quench in the clamp (x 2)
	ACR
	
	 Not resolved.



	Thursday 28_A
	TBE1. Communication problem in TEMA


	AB/CO
	
	 Equip.support called.

“Check communication test” launched.

	Wednesday 3_M
	TBE2 : IAP@warm unable to remove the trigger state in V_supra_ext and V_clamp _ext potaim cards


	MTM
	
	 Equip.support called.

VT1 5 lead in the CFB is open. Resolved by shorting VT1_4 with another tap.

	Wednesday 3_M
	TBE2: :IAP@warm : PLC interlock not OK. Magnet300_airX2 not OK. 
	ACR
	IAP
	Software implemented by A.Tovar not corresponding to the logic of IAP.

Problem solved.

	Thursday 4_M
	TBE1:LF on line and IAP did not show proper temperature through TT821 
	MTM
	IAP/LF
	Old version of IAP installed by equip.support.

	Thursday 4_M
	Problem for HV test for correctors (TBC1/3046) 
	MTM
	
	Done with the megger in the plate-form.

Non conformity open for 3046.

	Thursday 4_N
	TBE 2 : Intercon  cannot be configured properly.


	AB/CO
	
	Investigation of the communication between PLCSE and server.

Solved by AB/CO. New version replaced by the old version of the module of communication.

	Friday 5_M
	TBE 2 : TT821 reading not correct during  8 K instead of 50 K cool down 


	AB/CO
	
	Bad contact in the connector

	Friday 5_M
	TBC1: voltage does not correspond to the 2 A send to the magnet. 1.9 V instead of the 12 V (at warm)


	MTM
	
	Magnet with Left diode. Current flow through the diode.

Polarity of terminals power supply reversed.

	Friday 5_M
	TBA2: IAP @warm not started.

Two of the four quench heater power cannot be reset.


	MTM
	
	Two cards in the safety matrix not placed in the good slot. Re by piquet.

	Friday 5_N
	TBC1: Provoked quench at 1.5 kA (1 LF).

Faulty quench displayed by AQA


	MTM
	
	Test repeated. Then OK

	Saturday 6_A
	TBA2 (3028) : Leak in the manchette. Cool down did not start


	ACR/ICS
	
	Leak not localized.

Cool down started.

	Saturday 6_A
	TBE2 (1031) : MIITS  (36 )and Iearth too high (12 mA)


	MTM
	
	Test coordinator informed.

	Saturday 6_A
	TBE2 (1031) : Resistance of the cryo-heaters and temperature sensors too high.


	MTM
	
	Test coordinator informed.

	Saturday 6_A
	TBE2 (1031) : Lfweb.vi could not be seen in the screen and could not be re-launched from TEMA


	MTM
	
	TEMA re-started.

	Sunday 7_A
	TBC1 (3028) : 1.5 kA quenches (with HF) in Vbus.


	MTM
	ACR
	Not solved.

	Sunday 7_A
	TBE2:MM: over-range in sector of the heads (sector 1) in Ap1 and sector 0607 (ap 2).


	MTM
	
	Bad contact (Ap1) and pin removed for a connector in Ap2.

	Sunday 7_A
	TBE2:IAP at cold.

Offset errors in potaim cards and amplitude errors


	MTM
	
	Adjusted at the minimum possible.

New range to be defined?

	Sunday 7_A
	TBE2  Tcoil  . Problem with the connection of the terminals.


	MTM
	
	Temporary cabling present Friday.

Now fixed and definitive cabling. Solved by Giorgio.

	Sunday 7_N
	TBE2 Tcoil : MM current reading is zero in aperture 2 


	SL/PO
	
	Problem with one card in the power supply and one cable was not working.

	Monday 8_A
	TBE2:  MXI errors. At the start of the LHC cycle.


	AB/CO
	MMP
	Reset of the crates VME 


